Author's Response To Reviewer Comments

Clo<u>s</u>e

Dear Hans,

Please find attached our revised manuscript (in .docx format). We have added citations to "Availability of Data and Materials" section and to reference list. We also have removed highlighting in red and changed affiliation #11.

Yours Sincerely. also on behalf of other authors, prof. Yurii Aulchenko and dr. Yakov Tsepilov

Clo<u>s</u>e